Author’s response to reviews

Title: Eosinophilic Granulomatosis with Polyangiitis (Churg-Strauss Syndrome) Presenting as Diffuse Myositis

Authors:

Marc-Etienne Parent (Marc-Etienne.Parent@USherbrooke.ca)
Benjamin Ellezam (benjamin.ellezam@umontreal.ca)
Sandrine Larue (larue@neurorivesud.ca)

Version: 2 Date: 18 July 2014

Author’s response to reviews: see over
Eosinophilic Granulomatosis with Polyangiitis (Churg-Strauss Syndrome) Presenting as Diffuse Myositis

Authors:

- Marc-Etienne Parent, MD (Corresponding Author)
- Benjamin Ellezam, MD, PhD, FRCPC
- Sandrine Larue, MD, Bsc, FRCPC

The following changes have been made to the original document:

1. Line numbering has been added.
2. Abstract has been modified to adopt the desired structure.
3. Information on informed consent was added. The written consent had been obtained previously. The consent form for this specific journal has been sent to the patient and will follow shortly.
4. The format of the Case Report has been updated to comply with the usual structure used in this journal. Including Background, Conclusion and List of Abbreviations.
5. A competing interests section has been added.
6. An author’s contributions section has been added.
7. The figure has been removed for the manuscript.

Of note, Figure 1 is available as a high quality TIFF image (>100 MB) and can be sent to you on demand.

Thank you for your time and consideration,

Marc-Etienne Parent M.D.